Author's response to reviews

Title: Bioelectrical activity of the pelvic floor muscles during synchronous whole-body vibration - a randomized controlled study

Authors:

Magdalena Stania (m.stania@awf.katowice.pl)
Daria Chmielewska (d.chmielewska@awf.katowice.pl)
Krystyna Kwasna (k.kwasna@awf.katowice.pl)
Agnieszka Smykla (a.smykla@awf.katowice.pl)
Jakub Taradaj (j.taradaj@awf.katowice.pl)
Grzegorz Juras (g.juras@awf.katowice.pl)

Version: 6 Date: 16 September 2015

Author's response to reviews: see over
Dear Editors,

Thank you very much for your kind remarks regarding our manuscript entitled: “Bioelectrical activity of the pelvic floor muscles during synchronous whole-body vibration – a randomized controlled study”. Considering Editorial Board comments we have made the necessary additions:

1) Retrospective Trial Registration (no. ACTRN12615000966594); registration date: 15/09/2015)

2) Consort checklist.

The consort checklist has been filled in based on the content of the manuscript.

We have added the trial registration number and date at the end of the abstract.

We would like to emphasize that we present the results of a preliminary study and that we are aware of the resultant limitations. Nevertheless, the manuscript has received positive comments of two independent Reviewers and the results have been assessed to be interesting.

We hope that the revised manuscript conforms to the journal’s requirements and fulfills the criteria for publication in BMC Urology.

Yours faithfully,

Authors